Author’s response to reviews

Title: Evaluation of the level of cortisol, capillary blood glucose, and blood pressure in response to anxiety of patients rehabilitated with complete dentures

Authors:

M. C. Goiato (goiato@foa.unesp.br)
Emily Freitas da Silva (emilyvfs@yahoo.com.br)
Nádia Cândido (nadiabcandido@hotmail.com)
Adhara Nóbrega (adharasnobrega@gmail.com)
Rodrigo Antonio de Medeiros (rodrigomedeiros88@hotmail.com)
Doris Sumida (dorishs@foa.unesp.br)
Fernando Chiba (fernandoychiba@hotmail.com)
Daniela Micheline dos Santos (danielamicheline@foa.unesp.br)

Version: 2 Date: 10 Apr 2019

Author’s response to reviews:

We thank you for your attention and for your consideration concerning the manuscript entitled "Evaluation of the level of cortisol, capillary blood glucose, and blood pressure in response to anxiety of patients rehabilitated with complete dentures" (OHEA-D-19-00019R1). We are resubmitting that article with the corrections and accounting considerations of the reviewers, which can be found below.

We are willing to make further modifications and corrections, if necessary.

Best regards,

Marcelo Goiato
Corresponding author

Reviewer #1 Comments:
1. Figures 1 and 2 do not appear to be uploaded. Please upload them in one of the following formats:

- PDF (suitable for diagrams and/or images)
- Microsoft Word (suitable for diagrams and/or images, figures must be a single page)
- PowerPoint (suitable for diagrams and/or images, figures must be a single page)
- TIFF (suitable for images)
- JPEG (suitable for photographic images, less suitable for graphical images)
- PNG (suitable for images)
- BMP (suitable for images)
- CDX (ChemDraw - suitable for molecular structures)
- EPS (suitable for diagrams and/or images)

Response: Figures were uploaded in TIFF format.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We thank for your considerations. We are willing to make further modifications and corrections, if necessary.

We thank you for your attention and for your consideration concerning the manuscript entitled "Evaluation of the level of cortisol, capillary blood glucose, and blood pressure in response to anxiety of patients rehabilitated with complete dentures" (OHEA-D-19-00019R1). We are resubmitting that article with the corrections and accounting considerations of the reviewers, which can be found below.

We are willing to make further modifications and corrections, if necessary.

Best regards,

Marcelo Goiato
Reviewer #1 Comments:

1. Figures 1 and 2 do not appear to be uploaded. Please upload them in one of the following formats:

- PDF (suitable for diagrams and/or images)
- Microsoft Word (suitable for diagrams and/or images, figures must be a single page)
- PowerPoint (suitable for diagrams and/or images, figures must be a single page)
- TIFF (suitable for images)
- JPEG (suitable for photographic images, less suitable for graphical images)
- PNG (suitable for images)
- BMP (suitable for images)
- CDX (ChemDraw - suitable for molecular structures)
- EPS (suitable for diagrams and/or images)

Response: Figures were uploaded in TIFF format.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We thank for your considerations. We are willing to make further modifications and corrections, if necessary.